

Article

Integrating Biosignals Measurement in Virtual Reality Environments for Anxiety Detection

Livia Petrescu ¹^(b), Cătălin Petrescu ²^(b), Oana Mitruț ^{2,*}^(b), Gabriela Moise ³^(b), Alin Moldoveanu ²^(b), Florica Moldoveanu ²^(b) and Marius Leordeanu ²

- ¹ Faculty of Biology, University of Bucharest, 050095 Bucharest, Romania; livia.petrescu@bio.unibuc.ro
- ² Faculty of Automatic Control and Computers, University Politehnica of Bucharest, 060042 Bucharest, Romania; catalin.petrescu@acse.pub.ro (C.P.); alin.moldoveanu@cs.pub.ro (A.M.); florica.moldoveanu@cs.pub.ro (F.M.); marius.leordeanu@cs.pub.ro (M.L.)
- ³ Faculty of Letters and Sciences, Petroleum-Gas University of Ploiesti, 100680 Ploiesti, Romania; gmoise@upg-ploiesti.ro
- * Correspondence: oana.balan@cs.pub.ro

Received: 2 November 2020; Accepted: 9 December 2020; Published: 10 December 2020



Abstract: This paper proposes a protocol for the acquisition and processing of biophysical signals in virtual reality applications, particularly in phobia therapy experiments. This protocol aims to ensure that the measurement and processing phases are performed effectively, to obtain clean data that can be used to estimate the users' anxiety levels. The protocol has been designed after analyzing the experimental data of seven subjects who have been exposed to heights in a virtual reality environment. The subjects' level of anxiety has been estimated based on the real-time evaluation of a nonlinear function that has as parameters various features extracted from the biophysical signals. The highest classification accuracy was obtained using a combination of seven heart rate and electrodermal activity features in the time domain and frequency domain.

Keywords: virtual reality; biophysical; emotion recognition; experiment; EDA; human-computer interaction

1. Introduction

The integration of biophysical signals recordings in immersive virtual reality (VR) applications and experiments may significantly improve the results of psychotherapy, as physiological responses proved to be more useful than subjective reports in estimating users' emotions/anxiety levels during interactive sessions. By enhancing the human-machine interfaces and by augmenting the naturalness, fluidity, and intelligence of communication [1,2], we can facilitate the synchronization of physiological signals with VR scenario events to conduct a causality analysis.

Due to the complex and multidimensional nature of emotions, a holistic approach is needed to characterize the global psychophysiological responses using different evaluation systems. In the literature, there are two main methodological ways, namely: Techniques that use self-assessment questionnaires [3] and techniques that use measurements of various physiological parameters. These techniques can be used separately, but are often employed simultaneously to increase the reliability and quality of results [4]. The activity of the vegetative nervous system, responsible for the causality of emotions, is mostly evaluated by recording electrical impulses from the nervous system (electroencephalography—EEG) and by measuring various body parameters, such as electrodermal activity (EDA) and heart rate (HR).

The results of the simultaneous recording of these parameters can sometimes show variations in the same direction, but in other situations, they vary in opposite directions. These evaluation



techniques address different emotion dimensions—arousal, valence, or dominance [5]. For instance, EDA is a sensitive indicator for lower arousal range, responsible especially for cognitive processing, while HR is an indicator of higher arousal range, being related especially to the somatic mechanism of arousal [6].

As VR techniques started to be used in biomedical applications relatively recently, there are still no standardized protocols regarding the special conditions for the acquisition and processing of biological signals. The existing protocols cannot always be successfully used in VR applications, due to particularities, such as the long duration of the acquisition session, the large volume of resulting data, or the numerous artifacts caused by the subject's dynamics during the experiment.

This paper proposes a protocol, specifically designed to improve the biophysical signals' acquisition and processing quality, which may estimate the subject's anxiety level based on physiological measurements. This estimation can be used to adapt the phobia therapy sessions according to the subject's emotional state. For a better understanding of the protocol, we propose a case study analysis.

Section 2 presents emotions and exposure therapy alternatives, Section 3 describes the virtual reality applications used in psychophysical experiments, Section 4 discusses the most relevant biophysical signals-electrodermal activity, heart rate, and electroencephalography, Section 5 presents various protocols for physiological data acquisition in VR, Section 6 brings forward the proposed methodology for physiological data acquisition in VR, while Section 7 describes the case study for acrophobia therapy. Finally, the results, discussions, and conclusions are exposed in Sections 8–10.

2. Emotions and Exposure Therapy

Emotion is a mental state that appears spontaneously and without conscious effort, reflected by physiological changes in the brain, heart, muscle, and tissue activity. Emotions have been classified using two models: The discrete model, which divides emotions into various categories, and the continuous model, where emotions are labeled using multiple dimensions. The discrete models emerged considering the idea that emotions are innate, common in similar situations, and manifested psychologically and physiologically with comparable patterns across all people.

Initially, Ekman [7] identified six basic emotions: Happiness, sadness, anger, fear, surprise, and disgust and considered that the other affective states result as a combination of these six basic emotions. A set of eight basic emotions has been proposed: Joy, trust, fear, surprise, sadness, disgust, anger, and anticipation [8] and later, a set of ten basic emotions has emerged: Surprise, joy, interest, sadness, fear, shyness, disgust, guilt, anger, and contempt [9].

The continuous model classifies emotions based on the arousal and valence dimensions [10]. Valence describes the quality of the emotion and ranges from positive (pleasant) to negative (unpleasant). Arousal characterizes the degree of physical and cognitive stimulation and ranges from low arousal (passive) to high arousal (active) [11]. Emotions are spanned across this 2D emotion space model. As various, very different emotions fall within the same quadrant in the 2D model, a third dimension has been proposed by Mehrabian [12]. Dominance reflects the ability to control emotions and ranges from submissive to dominant. As an example, both fear and anger belong to the same quadrant in the 2D space—low valence and high arousal. What differentiates them is the fact that anger is a dominant state, while fear is submissive (low dominance).

Fear and anxiety are emotions that allow an appropriate response to a dangerous situation or threat. Although both are alarm signals, each prepares the body for a different kind of response. Thus, fear is a response to external, current danger, while anxiety is a response to an imagined threat, or an internal conflict. It is important to emphasize the indispensable cognitive component of the anxiety reaction. Consequently, anxiety can be seen as a more elaborate form of the fear reaction, allowing us to adapt to potential reactions and plan for the future [13]. Phobic-anxiety disorders are a group of manifestations characterized by exaggerated, uncontrollable, disproportionate fear, and avoidant behavior. Phobic people can present both intense mental discomfort that can lead to panic attacks and physical discomfort, with palpitations, breathing disorders, generalized weakness, profuse

sweating, dizziness, etc. It is important to make a difference between fear and phobia. Fear has an important adaptive role, being a transient state that appears in direct confrontation with danger, while a phobia is a pathological tonic state of preparation and continuous prediction [14].

VR exposure therapy has become an option for in vivo exposure and has several advantages: Authenticity of exposure, safe exposure, unlimited possibilities of training situations, high control of the variables for the clinician, and low costs. The field that examines the affective states is called affective computing. In 1997, Picard defined affective computing as: "Computing that relates to, arises from, or deliberately influence emotions" [15]. All the applications aimed at studying emotions and their psychological and physiological responses in education, healthcare, therapy, marketing, media, etc., are part of the broad field of affective computing.

3. Virtual Reality in Psychophysical Experiments

In psychophysical experiments, **emotion elicitation** has been performed using various stimulation materials: Pictures, music, short videos, games, virtual reality, self-induction methods. The International Affective Picture System (IAPS) [16] or the Chinese Affective Picture System (CAPS) [17] contain a set of standardized, open-access emotion-eliciting pictures available for a wide range of research experiments. Besides IAPS, the Center for the Study of Emotion and Attention from the University of Florida proposed the International Affective Digital Sounds [18] dataset, which contains auditory stimuli across a wide range of semantic categories [19]. The Affective Norms for English Words (ANEW) provides emotional ratings of valence, arousal, and dominance for a large number of English words [20]. The Affective Norms for English Text (ANET) contains emotional ratings of valence, arousal, and dominance for a large set of short English texts [21]. Music videos have been used for eliciting emotions in [22,23]. In [24], one actor tried to self-induce five major emotions: exhaltation, happiness, boredom, fear, and a neutral state.

The modalities of quantifying emotions are: Self-reports (verbal descriptions of the emotions experienced, self-ratings using the Self-Assessment Manikins (SAM)—various anthropomorphic figures [25] depicting different emotions, the Affective Slider [26]—a valid alternative to Self-Assessment Manikins, observations of the human behavior and measures of the physiological signals originating in the Central Nervous System (brain activity collected by EEG) and Autonomic Nervous System (respiration, heart rate, body temperature).

Virtual reality creates a complete sense of presence and increases immersivity by using stereoscopic 3D view, motion tracking, and head tracking technology [2]. It is a rapidly growing field, expected to rise to 80 billion dollars by 2025 [27]. Immersive Virtual Reality is able to trigger emotional states, quantified by various physiological responses.

Spatial presence is an important prerequisite of any VR application, describing the perception of being present in the virtual environment [28]. Spatial presence enhances game attractivity and improves players' performance [29]. Felnhofer et al. [30] demonstrated that the sense of presence is linked to emotional responses in an experiment where electrodermal activity has been recorded.

The feeling of presence in the virtual environment, the degree of realism, and immersivity are highly correlated with the percentage of change in heart rate, and skin resistance. Wiederhold [31] found a significant correlation between presence and skin conductance level in VR. Riva et al. [32] found that the sense of presence influences the emotions experienced in the virtual environment. Jang et al. [33] found that different virtual environments generate different affective states. Moreover, electrodermal activity and heart rate variability are reliable measures of arousal. In Mehan et al. [34], the intensity perceived in the virtual environment was directly proportional to the heart rate. Peterson et al. [35] showed that high virtual height conditions increased heart rate variability, electrodermal activity, and heart rate frequency power compared to low virtual heights, inducing a certain level of psychological stress. Virtual characters, such as avatars, are capable of eliciting emotional responses in people who observe their facial expressions and postural gestures [36–38].

Virtual reality has been successfully used for phobia therapy. In [31], the post-therapy behavior of 33 phobic subjects who have been treated for fear of flying showed a gradual trend towards the nonphobics' physiological responses, due to desensitization (the process that diminishes emotional responsiveness to negative or aversive stimuli) and habituation (the repeated presentation of identical, non-significant stimuli which elicits a decreasing response intensity with progressively smaller reactions). VR is efficient for treating post-traumatic stress disorder [39] and for military training [40]. The affective virtual reality system (AVRS) [41] contains eight VR scenes, rated in terms of valence, arousal, and dominance by 100 subjects.

Although VR is very attractive, there are some limitations, such as the cost of the VR devices, the computing power required by the VR software, motion sickness, size and weight of the head-mounted displays, graphical limitations, potential addiction or eye damage caused by the VR lenses are placed very close to the eyes [42].

4. Biophysical Signals

4.1. Electrodermal Activity

Electrodermal activity (EDA) or Galvanic Skin Response (GSR) measures the changes in conductance at the skin surface due to eccrine sweat production. EDA is a good biomarker of autonomic system response, being a "window on the arousal dimension of emotion" [43]. The eccrine glands are the main sweat glands in the body being active in the biological adaptative response, by thermal and emotional sweating. With a total number of approximately 2 to 5 million, the eccrine glands have the highest density in the palm and in the soles, then on the head, but much less on the trunk and in the extremities. Males and females have the same number of sweat glands, the size, and volume of the sweat secreted by each gland, being about five times greater in males [44]. EDA reflects only activity within the sympathetic branch of the Autonomic Nervous System (ANS), being involved in emotional-induced sweating in response to various stimuli [45,46]. Modern functional brain imaging and neurophysiological techniques [43]. have demonstrated that the EDA mechanism is highly modulated within the limbic system via the hypothalamus and the thermoregulatory pathways (for orientation and defensive responses) and to a lesser degree by the premotor cortex and the basal ganglia (involved in the control of limb movements) [47]. Consequently, EDA recordings allow a non-invasive functional assessment of gross movements, attention, emotion perception, threat responses, affective and motivational processes, orientation, decision-making, and fine control [48,49].

EDA continues to be used in psycho-behavioral studies, allowing the assessment of brain activity and autonomic arousal. It is a valuable tool, as it is generally difficult or impossible to influence or control the skin response to emotional stimuli [50]. Figure 1 presents the diagram of the EDA non-stationary signal that decomposes the response to specific tasks into two quantitative measures: *The tonic component*—Skin Conductance Level (SCL) and *the phasic component*—Skin Conductance Responses (SCR), which needs to be evaluated separately [51].



Figure 1. Diagram of the Electrodermal activity signal decomposition.

The tonic component is not associated with the onset of a specific stimulus. It is a measure of slow changes in EDA (minutes instead of seconds) and background characteristics of the signal, which specifically refer to the "level" of skin conductance, usually expressed in microSiemens, μ S.

The changes in the SCL reflect general changes in autonomic arousal, such as emotional state and stress level [52]. The tonic level of skin conductance can vary slowly over time in a person, depending on age, psychological state, hydration, and self-regulation.

The *phasic component* is associated with a specific and identifiable stimulus, being a measure of rapid changes in EDA (measured in seconds) that result from a momentary sympathetic activation. It is referred to as the "response" of skin conductance—SCR [53].

Mean SCL values usually range between 2 and 20 μ S. Maximum SCR amplitudes vary within 1 and 3 μ S [6]. The Event-Related SCR (ER-SCR) is the sensitive component to emotionally specific events. After a latency period of about 1–3 s (with an average value of 1.8 s in optimal environmental conditions), SCR shows a steep onset, with a typical rise time of about 1 to 3 s, reaching an amplitude of at least 0.1 μ S, a flatter recovery and an exponential decay with a recovery time of half of the SCR amplitude between 2 and 10 s, as can be seen in Figure 2. A minimum of 0.05 or 0.04 μ S is typically set as a threshold to define a significant SCR [51].



Figure 2. Event-related skin conductance responses SCR.

Spontaneous fluctuations in electrodermal responses are known as Non-Specific Skin Conductance Responses (NS-SCRs). They are phasic increases in EDA that have the same appearance as stimulus-elicited EDA responses, but because they are not associated with any specific stimuli or artifacts, they are considered to belong to the tonic component of skin conductance [51]. NS-SCR appears 5 s after the stimulus finished [6]. The typical frequency of NS-SCR is 1–3 per minute during resting periods and over 20 per minute in high arousal situations. An increased frequency of NS-SCRs is considered to be a biomarker of stress, emotions, and anxiety.

Habituation was considered to be a basic way of learning, and SCR was a frequently used indicator for it. [47].

Experimental research has concluded that some people have special electrodermal signal characteristics, similar to a signature, that is influenced by both genetic and environmental factors [54]. A certain category of people has a very small variation of the signal when they are stimulated, compared to the resting state, with fewer NS-SCRs and a more rapid specific EDA habituation. They are classified as "stable individuals". Opposed, there is the population category with increased reactivity to stimuli, with a higher rate of NS-SCRs and slow habituation to specific stimuli. They are classified as "labile individuals" [55]. There have been attempts to associate these types of electrodermal responses with certain personality traits [47]. People described as calm, deliberative, restrained, good-natured, cooperative, and responsible are the EDA-labiles, while people who are active, emotionally expressive, animated, assertive, more irritable, antagonistic, impulsive, and irresponsible are EDA-stabiles. Some individuals who show electrodermal hypoactivity can have affective disorders with vulnerability for depression or suicide attempts [47,56].

The phobic reaction is a defense-like overreaction to specific stimuli [6]. Several studies have shown that phobic patients do not show a general increase in electrodermal activity, but there is an overreaction only to specific stimuli. The increased amplitude of SCR may be a sensitive indicator for fear responses within the phobia condition. SCL and NS-SCR increased significantly during phobic stimuli presentation [57,58]. The results obtained by Herbelin et al. [24] suggested that SCL and the standard deviation of the heart rate are relevant for assessing emotion intensity. In [31], the percentage change of skin resistance showed significant differences between the nonphobics and phobics subjects

6 of 31

after a VR therapy for fear of flying. The phasic component was revealed when arousing stimuli has been presented, although they also occurred spontaneously in some individuals [59–61].

4.2. Heart Rate

The basic heart rate (HR) is provided by the cardiac conduction system, a network of specialized cardiac muscle cells that ensures the *intrinsic innervation* of the heart and that spontaneously initiate and transmit the electrical impulses responsible for the coordinated contractions of each cardiac cycle. The sinoatrial node, part of the cardiac conduction system (cardiac pacemaker), has its own frequency that varies between 95 and 110 beats per minute, being dependent on age. The ANS ensures the extrinsic innervation of the heart, modifying the activity of the intrinsic system, according to the needs of the organism. The resting heart rate is within the range of 60–70 beats/min, with obvious domination of parasympathetic influences on the sinus node [62]. Sympathetic influences increase the pulsation rate. Assessment of heart rate variability (HRV) is a non-invasive method that actually describes the oscillations between consecutive R-R intervals of the electrocardiogram, as a result of the modulatory effect of the ANS and of catecholamines circulating in blood on cardiac activity. This fluctuation, known as the HRV index, is an indicator of the body's ability to react to the variations of the external and internal environment [63,64].

The vagus nerve, parasympathetically innervates many of the body's organs, being responsible for most of the extrinsic innervation of the heart. Distributed especially to the nodules of the cardiac conduction system, the vagus actively attenuates the influences of the sympathetic nervous system (SNS) on the heart. HRV at rest is indicative of ANS health and higher emotional well-being, HR reactivity being a change from the basal state in response to stimuli. When an individual is threatened, the vagal tone is inhibited, and SNS is activated, triggering a suite of responses that promote survival, which affects the decrease of HRV. Polyvagal theory [65] associates higher levels of HRV with the capacity for social engagement, and the neurovisceral integration model (NIM) [66] postulates that higher resting HRV indicates tendencies for appraisal [67]. High HRV, (the parasympathetic influence on heart rate), is associated with a range of physiological processes, adaptive responses, and good adaptability to changes [68]. Related to the emotional state, high HRV is correlated with low anxiety and reduced incidence of phobias, low rumination, and a low level of worry [69]. Low HRV at rest, characterizing the SNS activity on heart rate, tend to be less flexible both physiologically and behaviorally in adapting to environmental demands [70]. Low HRV at rest was associated with difficulties in emotional regulation [71] and psychiatric conditions (depression, anxiety, and alcohol use disorders) [72,73] Very low HRV is conditioned by the humoral influence on HR [62].

4.3. Electroencephalography (EEG)

EEG is an electrophysiological method of recording the electrical activity of the brain and is done by attaching electrodes to the scalp, according to a standard format. The various patterns of electrical activity are called waves or brain oscillations and can be identified by amplitude and frequency. Each category of oscillations is associated with certain mental states. For example, the alpha waves (8–13 Hz) are related to states of relaxation, meditation, or mental rest [74]. The EEG measurements are usually based on contrasting the activations of fairly large regions of the brain [75,76]. Initial studies related frontal asymmetry to emotional valence, the left hemisphere is associated with positive emotions, and the right hemisphere with negative emotions. Subjects with higher left frontal activity were reported higher levels of general well-being [77], as they had stronger positive affective reactions to videos with more pleasant content and negative reactions less intense in those with aversive content [78]. Subsequent studies have shown that the method of frontal asymmetry reflects the balance between tendencies of proximity and distance rather than emotional valence [79].

5. Protocols for Physiological Data Acquisition in VR

PhysioVR [2] is an open-source framework that facilitates the integration of physiological signals in mobile VR applications. It enables connectivity with the Unity 3D game engine, external event triggering and acquisition, streaming and recording of physiological signals. PhysioVR is composed of two software layers: PhysioSense, which synchronizes the devices and streams data, and PhysioAdapt, which receives and adapts the biophysical signals in the Unity 3D game. By using the Wizard of Oz methodology [80], stimuli can be created, and events can be triggered in real-time. Data flows in real-time between the devices, and the physiological signals can be plotted in real-time along with their corresponding stimuli or events. The protocol has been tested using a VR game called EmoCat Rescue, where the players were required to find a lost cat by listening to stereoscopic sounds. The application offered real-time feedback on the cardiac rhythm, encouraging messages and indications about carrying out breathing exercises that would reduce irregular heartbeats.

Rehabnet Control Panel [81] enables interfacing of physiological sensors with VR applications using the Virtual-Reality Peripheral Network (VRPN) and User Datagram Protocol (UDP). It allows the recording of biophysical data, game events, and visual stimuli.

6. Proposed Methodology for Physiological Data Acquisition in VR

The increasing use of VR immersion technology for various applications, especially for detecting anxiety in the biomedical field, requires standardized protocols to address the special steps for data collection and management. The measurement of physiological parameters, objective indicators of the level of anxiety, are unique in VR experiments, because of the particularities of this type of application. We can mention here the need for preparatory steps for: Familiarizing the subject with the equipment, obtaining a baseline to remove interindividual variability, identifying irresponsive people, efficient signal filtering to eliminate power grid noise, etc.

Additional hardware considerations, the need to evaluate anxiety in real-time for the correct guidance of the therapy session, the long duration of the acquisition session, the large volume of resulting data, or the numerous artifacts caused by the subject's dynamics during the experiment are some of the particular restrictions that occur in this type of experiment.

In this article, we propose a protocol that takes into account the most common of the particular constraints that occur in this type of experiment. At the same time, this paper provides a series of recommendations and suggestions to minimize these shortcomings and to draw up a checklist of standardization procedures for data collection and management, with a focus on assessing anxiety.

Figure 3 presents the diagram of the proposed experimental design and protocol for emotion/anxiety detection in VR.

6.1. Preparation of the Experiment

Before carrying out the experiment, a stage of preparation is indicated. This stage has to be scheduled before the experiment, not mandatorily in the same day. Certain aspects must be taken into account:

- Obtaining informed consent from the participants after explaining the aim and conditions of the
 experiment helps ensure their well-being and maintains high ethical standards.
- Filling in a demographic questionnaire specifying age, race, ethnicity, gender, marital status, income, education, employment, health status, previous experience in the virtual environment gives a lot of information about the subjects and allows for the analysis of statistical correlations. In the case of VR applications for emotional evaluation in phobia, anxiety, stress, etc., a short psychometric evaluation is required by applying dedicated questionnaires. They are designed to assess the characteristic symptoms and severity of emotional impairment, using subscales as fear, avoidance, physiological arousal to obtain a total score.

- A preliminary test or a pilot study, to familiarize the subject with the experimental device (especially in the case of VR equipment) will avoid the stress induced by the impact with the unknown devices. This will help the researchers to find out how long the participants felt comfortable while watching immersive videos before they experienced fatigue or motion sickness [82].
- It is also necessary to verify and classify individual reactivity, followed by excluding individuals without reaction, because for these individuals, the EDA analysis does not provide too much information. The evaluation of individual reactivity can be done by evaluating the intensity of the physiological responses during the subjects' exposure to images with known emotional value. About 10% of the participants can be non-responders [52]. This stage can also provide information about the "unstable" or "stable" nature of the subjects' physiological response.



Figure 3. Block diagram of the experimental design and protocol for emotion/anxiety detection in virtual reality (VR).

6.2. Data Acquisition

In order to ensure that quality data is acquired both in the stage of preparation, and especially during the experiment, two important conditions should be fulfilled:

- The design of an appropriate environmental setup, by creating a relaxing atmosphere for the participants that includes a quiet, noise-free environment (with a minimum circulation of people), with an optimal ambient light that ensures emotional balance and reduced unwanted external stimuli. Maintaining the temperature at a constant value (around 23 °C) and the humidity between 40 and 60 ensures that the body functions at the optimum metabolic rate. The period of day (morning, afternoon, or night) should be standardized within the same research project, considering the possible influences of the circadian rhythms [83].
- The setup of the recording device and the placement of the Ag/AgCl electrodes should be done correctly. There are several configurations for their positioning, which involves the regions where the eccrine sweat glands have a higher density: The fingers, palms, or soles. In VR applications, mostly finger electrode placement is used. The sensors are placed on the index and ring finger for better cable management. It is recommended to place the electrodes on the adjacent fingers of the non-dominant hand (second and third fingers, or fourth and fifth fingers), because they are innervated by the same spinal nerve (median respectively ulnar) [84]. Due to the greater number of sweat glands, it is recommended to fix the electrodes on the distal phalanx of the fingers. Another important aspect at this stage is to avoid data recording errors resulting from the improper connection of the electrodes (either by faulty locations or by the lack of contact with the skin) and to remove possible electromagnetic interference. This requires cleaning the skin before

placing the electrodes (it is recommended to clean only with water), using high-quality electrodes, fixing them very well, and in some situations, even using electroconductive gel in contact with the skin. Before beginning the recording sessions, it is necessary to stabilize the skin-electrodes interface. The researcher has to *check the sensors' sensitivity*, by asking the participant to take a deep breath in and to hold it for a few seconds. A good EDA signal should contain a spike in skin conductance within 2–3 s after the breath was initiated [51].

During exposure therapy, the physiological parameters can be monitored by techniques such as EDA, HR, EEG, etc. For the daily practice of a psychotherapist, the use of EEG is more difficult, the installation of electrodes being time-consuming and introducing an additional source of stress. The VR device can be an obstacle to the correct placement and stability of the electrodes. Therefore, in the proposed protocol, we will focus mainly on EDA and HR.

- Baseline recording. At the beginning of the experiment, once the physiological sensors and VR equipment are placed, it is advised that the subject should wait several minutes (typically 5–15 min) before beginning the recording session, in order to stabilize the skin-electrode interface [53]. Once the interface was stabilized, to obtain correct baseline values, the participants should be asked to relax for 60 s, without ample movements, without breathing deep, while watching neutral images. In this way, the non-specific signals are reduced. At the end of the baseline recording, it is indicated to apply a stimulus to obtain a reference response—for example, during a deep breath.
- VR exposure represents the actual section of therapy, which consists of the active participation of
 the subject in a scenario that involves the confrontation with phobic stimuli. The sensors acquire
 biophysical data, and analysis should be performed to highlight the relation between occurring
 events and the measured data. To maximize the therapeutic effect, the scenario should be adapted
 to the subject's condition, assessed through the degree of anxiety. The adaptation of the scenario
 aims to gradually exposure of the subject to the stimuli, thus avoiding unwanted situations of
 overreactions (panic attacks). Therefore, the real-time assessment of anxiety levels is mandatory.

6.3. Data Processing

6.3.1. Data Preprocessing

Raw biophysical data is preprocessed by following the next steps:

Data filtering refers to removing unwanted components from a signal. The filter is used to smooth noisy signals and signals unrelated to a specific analysis [85]. There are more filters that can be used in data processing: Basic median filters, low-pass filtering, exponential smoothing, Kalman filter [86]. By analyzing the EDA signals acquired during various experiments, we found that the noise with the highest amplitude is the one induced by the electrical network. The main component of this type of noise has a frequency of 50 Hz, but it may also contain significant harmonic components generated by non-sinusoidal currents of nonlinear loads. The high frequency of these components, relative to the spectrum of the EDA signal, allows their efficient filtering by using a low-pass filter. However, it is very important to use a proper sampling frequency in order to avoid the aliasing effect. According to the Nyquist–Shannon sampling theorem, the minimum sampling frequency must be 100 Hz, to avoid aliasing at the fundamental frequency, but it is advisable to choose a higher frequency due to the presence of the harmonic components. The first step in implementing the low-pass filter is to decide between the two main categories of filters: Infinite Impulse Response (IIR) and Finite Impulse Response (FIR) filters. Although the IIR filters are much more efficient in terms of computational effort, required memory, and have a shorter response time, we recommend using FIR filters because they have a shorter transition band, are much more stable numerically, and introduce fewer shape distortions of the signals due to their linear phase characteristic. The heart rate information taken from the data acquisition devices is the result of complex processing (that includes filtering), performed on electrocardiogram (ECG) or photoplethysmographic (PPG) signals. For this reason, additional filtering of the HR signal is often not necessary. The noise in the EEG signals is issued by eye blinks and body movements. In addition to the recommendation to minimize participants' movements, more techniques, such as least mean square and blind source separation, are used. Moreover, filter techniques are applied: 4 Hz, 5Hz, or even lower frequencies such as 0.5 Hz, 1 Hz as the lower-band cut-off frequencies; 30 Hz, 70 Hz as the upper-band cut-off frequencies. For the database construction, a 5th order Butterworth band-pass filter, 4 Hz lower-band, and 45 Hz upper-band filters are applied [87].

- The downsampling of a data sequence is the process of reducing the sampling rate [50]. This process can be achieved in multiple ways. The simplest of them is the sampling rate reduction by an integer factor, known as the "decimation" method. Decimation with an integer factor n consists of taking a sample from a data sequence and discarding n 1 samples for every n samples [50]. Downsampling is usually required when the signal's sampling rate is higher than the one imposed by the bandwidth of the signal. In the case of the EDA signal, the efficient elimination of the noise generated by the electrical network requires the use of such a high sampling frequency. Because further processing of the EDA signal requires a computational effort that depends heavily on the number of samples, downsampling is a necessary condition for ensuring the real-time processing of this signal. In order to avoid the aliasing effects caused by downsampling, it is necessary to reduce the sampling rate down to at least double the low-pass filter frequency limit.
- Normalizations and standardization refer to data transformation to perform good statistical data analysis. Normalizations means rescaling the values to fit in a range, including data transformation to correct for skew/kurtosis so that the data is fit for parametric statistical analysis (min-max, Z-score, log transformation, or square-root transformation). Standardization refers to another data scaling technique, to perform directly and meaningfully a comparison between subjects. Standardization may not always be necessary, but the most common data transformations include: Transformations into standard values (min-max, Z-score), Range-Corrected Scores, Proportion of Maximal Response, etc. [52,87].

6.3.2. Data Analysis

- Deconvolution. Skin conductance data is characterized by the overlap of two components: A slowly, continuously varying tonic component (skin conductance level—SCL) over which a rapid component overlaps (phasic skin conductance response—SCR), in close correspondence with the arousal states, whose amplitudes give information on the intensity of the emotional states. While data acquisition can be made quite easily, data analysis is a complex process, through which the discrete reactions superimposed over the tonic component must be decomposed. The variability of SCR shapes hereby complicates the proper decomposition of SC data. Benedek [88,89] proposes a method for the full decomposition of the SC data into tonic and phasic components, starting from the biological model of sweat diffusion, a model with two compartments. Two EDA analysis strategies are proposed: Continuous Decomposition Analysis (CDA) and Discrete Decomposition Analysis (DDA). CDA [88] is the recommended method for the analysis of skin conductance data. This method extracts the phasic information underlying EDA, deconvolving the SC data by the general response shape. The results of a large increase of temporal precision, the data is decomposed into continuous phasic and tonic components, with several standard measures of phasic EDA. DDA [89] uses non-negative deconvolution to decompose SC data. It is an algorithm difficult to use for real-time estimates, because it may be slow for large data samples.
- Features extraction is an essential step for emotion classification. Starting from an initial data set, after processing the signals, a multitude of derived data, called features, can be extracted and used to estimate the anxiety level. Before extracting the features, a segmentation stage of the collected data is recommended, by extracting and analyzing parts of the raw physiological signal, called time-windows. The length of the window can be different: The same as the stimuli length, shorter or longer than stimuli length. Not all the features have the same relevance for estimating anxiety,

and previous works have made limited contributions to identifying the most appropriate ones. In our study, we focus only on the features extracted from the skin conductance and heart rate related data, and we will try to identify the most relevant ones for assessing the level of anxiety in real-time.

6.4. Classification

The last step in this protocol involves the use of the features extracted from the measured physiological parameters in order to obtain an estimate of the anxiety level. Subsequent use of this level of anxiety to adapt to the scenario of VR exposure therapy can be facilitated by classifying it into several intensity categories according to an evaluation scale. Among the algorithms that obtained a high classification accuracy are the classical machine learning methods and the convolutional neural networks (CNNs) [90,91].

7. Case Study for Acrophobia Therapy

The aim of the case study consists, first of all, in validating the data acquisition protocol presented above. Moreover, we want to determine the physiological parameters that have the highest degree of correlation with the level of anxiety, which will be further used for estimating in real-time the subject's emotional states.

7.1. Description of the Experiment

An experiment was performed to expose the subjects to stimuli that can trigger the fear of heights (acrophobia), the exposure being achieved by immersion in VR.

A group of seven subjects, aged 22–50 years, was selected for the experiment. After completing the Visual Height Intolerance questionnaire [92] that assessed the degree of anxiety during heights exposure, the subjects have been classified into three categories: Two suffered from a mild form of acrophobia, three from a medium-intensity fear of heights, and two experienced a severe form of height intolerance. More details can be found in [93–95]. To familiarize the subjects with the experimental device and with the VR perception, we provided some VR introductory sessions. Before starting the experiment, all the subjects were informed about the experiment and signed informed consent. The experiment was approved by the ethics committee of the UEFISCDI project 1/2018 and UPB CRC Research Grant 2017, and University POLITEHNICA of Bucharest, Faculty of Automatic Control and Computers.

The experiment was designed as a game with visual and vestibular stimuli, rendered on the HTC Vive head-mounted display [96], in which the subjects had to collect 15 coins with different values (bronze, silver, and gold), situated on the ground floor and on the balconies of a building (at the first, fourth and sixth floor). Each time the user collects one coin, a virtual panel appears and asks the participant to rate the level of fear perceived on a gradual 11-choice scale called Subjective Unit of Distress (SUD) (0 for complete relaxation, 1–3 for low fear, 4–7 for medium stress level and 8–10 for high anxiety) (Figure 4).



Figure 4. The VR acrophobia game.

In order to evaluate the reliability and internal consistency of the self-reported anxiety levels, we performed a Cronbach alpha test (using IBM SPSS[®] Statistics software version 27.0). The resulted alpha coefficient was 0.91, which suggests that the self-reported data to have relatively high internal consistency.

Throughout the whole game, peripheral physiological data were recorded via a Shimmers Multi-Sensory device [97]. The Shimmer3 GSR+ Unit measures the skin conductance in microSiemens, GSR data being acquired at a rate of 51.2 samples per second and uses an optical pulse probe to capture a PPG signal that is later converted to estimate heart rate (HR). The data are sent via Bluetooth connectivity to a computer. Each time a coin is collected in the game, a timestamp is added to the file in which the recorded physiological data is saved.

During some experiments performed at the beginning of this study, we found the presence in the EDA signals of some periodic disturbances with amplitude and a period close to that of the electrodermal responses. Figure 5 shows a segment of the noise-affected EDA signal corresponding to a baseline recording.



Figure 5. Example of a signal affected by electrical network noise (**a**) signal in the time domain; (**b**) frequency spectrum of the signal.

The noise signal is generated by the electrical network, although, at first sight, the fundamental frequency of this signal (1.168 Hz) would exclude this cause. In reality, the recorded signal is the result of the aliasing effect introduced by incorrect signal sampling.

Because the sampling frequency used to record the signal was 51.2 Hz, the alias mechanism, shown in Figure 6, explains the presence of components with frequencies of 1.168 Hz, 2.337 Hz, ..., in the spectrum of the acquired signal as aliases of a fundamental frequency of 50.032 Hz and its harmonics.



Figure 6. Alias effect during sampling of the electrical network noise with a sampling frequency of 51.2 Hz.

In order to avoid this phenomenon, we recommended in the protocol the use of a sampling frequency much higher than twice the frequency of the noise generated by the electrical network, low-pass filtering to eliminate this noise, and downsampling of the filtered signal to return to a lower sampling rate compatible with the bandwidth of the EDA signal.

The data collected was processed to estimate the degree of anxiety reached by the participants during the game. At the same time, the predictive accuracy of the proposed protocol was evaluated.

7.2. Data Analysis

Data analysis has consisted of several steps: The deconvolution of the GSR signal, data segmentation, features extraction, the selection of the most informative features, and the determination of a correlation between the extracted features and the anxiety levels.

7.2.1. Deconvolution

The deconvolution of the GSR signal has been processed using the Continuous Decomposition Analysis algorithm [88] implemented in Ledalab V3.4.9 [98]. This algorithm extracts the tonic (SCL) and phasic (SCR) signal components and also identifies the amplitude (AMP) and onset time (TON) of electrodermal responses.

7.2.2. Data Segmentation

A data window of 5 s before the moment when the subject picked up a coin was extracted from all signals. This window corresponds to the maximum exposure to the phobic stimulus. For each time segment, we had a self-evaluation rating of the anxiety level between 0 and 10, because the subject was asked to rate his/her anxiety level after picking up each coin.

7.2.3. Features Extraction

As we have seen from the literature presented, a wide range of features can be extracted from raw data. But not all the parameters that can be extracted have the same importance in interpreting the biological significance of the data. Therefore, identifying the most relevant features or finding an optimal set of features that provide fast and accurate information on estimating anxiety is of major importance, especially when processing experimental data in real-time.

Thus, in our experiment, starting from the raw data, an extended set of features has been extracted. In this stage of the study, we selected all the features, which were suggested more frequently in the literature. In the next stage, following the feature extraction, an exhaustive search will be performed to select the most relevant features for anxiety level estimation.

The extended set of features contains three categories of features that characterize the biophysical signals in terms of time evolution, GSR response events, and frequency content as following:

Electrodermal Activity—Time Domain Features

The time domain features represent a series of parameters that are easily calculated, being descriptors of the waveform amplitude, frequency, and duration [99]. The signals are defined by their samples: $\{x_K\}_{K=0...N-1}$, where X_k is the *k*-th sample of the signal X, and N represents the number of signal samples inside the time segment.

(1) MAV— mean absolute value

$$X_MAV = \frac{1}{N} \sum_{K=0}^{N-1} |x_K|,$$
(1)

$$X_MAR = \frac{1}{N X_B} \sum_{K=0}^{N-1} |x_K|,$$
(2)

(3) STD—standard deviation

$$X_STD = \sqrt{\frac{1}{N-1} \sum_{K=0}^{N-1} (x_K - \bar{x})^2},$$
(3)

where $\overline{x} = \frac{1}{N} \sum_{K=0}^{N-1} x_K$ represent the mean value of the signal.

(4) WL—waveform length [100]

$$X_WL = \sum_{K=1}^{N-1} (\Delta x_K)^2,$$
 (4)

where $\Delta x_K = x_K - x_{K-1}$.

(5) SSC—slope sign changes

$$X_SSC = \sum_{K=1}^{N-2} f(x_{K-1}, x_K, x_{K+1}),$$
(5)

where
$$f(x_{K-1}, x_K, x_{K+1}) = \begin{cases} 1 \text{ if sign } (x_K - x_{K-1}) \cdot \text{sign } (x_{K+1} - x_K) < 0 \text{ and} \\ |x_K - x_{K-1}| \ge \varepsilon \text{ and} \\ |x_{K+1} - x_K| \ge \varepsilon \\ 0 \text{ otherwise} \end{cases} \text{ and } \varepsilon = 0.001 \text{ } \mu\text{S}.$$

(6) *WAMP—Willison amplitude*

$$X_SSC = \sum_{K=1}^{N-1} f(x_{K-1}, x_K),$$
(6)

where
$$f(x_{K-1}, x_K) = \begin{cases} 1 \ if |x_K - x_{K-1}| \ge \varepsilon_W \\ 0 \ otherwise \end{cases}$$
 and $\varepsilon_W = 0.5 \ \mu S.$

Electrodermal Activity-Events Related Features

(1) *NR—number of responses*—represents the number of electrodermal responses detected inside the analyzed time segment

$$X_NR = length (AMP) = length (TON),$$
(7)

(2) AVRA—average response amplitude

$$AVRA = \frac{1}{NR} \sum_{K=0}^{NR-1} AMP_K,$$
(8)

(3) MAXRA—maximum response amplitude

$$MAXRA = \max_{K=0...NR-1} (AMP_K),$$
(9)

Electrodermal Activity—Frequency Domain Features

The calculation of these parameters is a more laborious process because it requires evaluating the power spectrum density (PSD) of the electrodermal signals (GSR, SCL and SCR). For each data window, PSD is evaluated for a set of frequencies $\{f_K\}_{K=0...N-1}$ between 0 and 25.6 Hz with a 0.05 Hz step.

(1) *FMD—frequency median* [99] represents the frequency that divides the signal spectrum in two sections that have the same energy and is located in the frequency set used for the PSD calculation at the index *M* which minimizes the value:

$$\left|\sum_{K=0}^{M-1} P_K - \sum_{K=M}^{N-1} P_K\right|,$$
(10)

(2) *FMN—frequency mean* represents the average frequency weighted by the power spectrum density:

$$f_{MN} = \frac{\sum_{K=0}^{N-1} f_K P_K}{\sum_{K=0}^{N-1} P_K},$$
(11)

Heart Rate—Time Domain Features

Heart rate is defined as the time interval between successive heartbeats. For each data, window corresponds a vector $\{h_K\}_{K=0...P-1}$ that contains the heart rate values.

(1) *MAV*— mean absolute value

$$HR_MAV = \frac{1}{N} \sum_{K=0}^{P-1} |h_K|,$$
(12)

(2) *STD—standard deviation* is a measure of heart rate variability and is defined as:

$$HR_STD = \sqrt{\frac{1}{N-1} \sum_{K=0}^{P-1} (h_K - \bar{h})^2},$$
(13)

where $\overline{h} = \frac{1}{N} \sum_{K=0}^{P-1} k_K$ represent the mean value of the signal.

Heart Rate—Frequency Domain Features

In order to obtain this class of features, the heart rate signal ${h_K}_{K=0...P-1}$ is decomposed in two signals ${h_K}_{K=0...P-1}$ and ${h_K}_{K=0...P-1}$ that contain the frequency components between 0.04 Hz and 0.15 Hz, respectively, between 0.15 Hz and 0.4 Hz [101]. The signal decomposition is performed using two 3rd order band-pass Butterworth filters, as we can see in Figure 7.



Figure 7. Frequency response of decomposition filters.

(1) *LFI—low frequency signal intensity* represents the intensity of the signal containing frequency components between 0.04 Hz and 0.15 Hz:

$$HR_LFI = \sqrt{\frac{1}{P} \sum_{K=0}^{P-1} h l_{K'}^2}$$
(14)

(2) *HFI—high frequency signal intensity* represents the intensity of the signal containing frequency components between 0.15 Hz and 0.4 Hz:

$$HR_{HFI} = \sqrt{\frac{1}{P} \sum_{K=0}^{P-1} h h_{K'}^2}$$
(15)

(3) *HLR—high/low frequency intensity ratio* represents the ratio between intensities of low frequency and high frequency HR signals:

$$HR_HLR = \frac{HR_HFI}{HR_LFI},$$
(16)

For each data window of 5 s, the following features were extracted from the electrodermal and heart rate signals as presented in Table 1.

Physiological Parameter	Domain	Features Type	Features
		(1) MAV	GSR_mav, SCL_mav, SCR_mav
		(2) MAR	GSR_mav, SCL_mav, SCR_mav
	Time domain	(3) STD	GSR_mav, SCL_mav, SCR_mav
	Time domain	(4) WL	GSR_mav, SCL_mav, SCR_mav
		(5) SSC	GSR_mav, SCL_mav, SCR_mav
EDA		$Time domain = \begin{cases} (1) & MAV & GSR_mav, SCL_mav, SC \\ (2) & MAR & GSR_mav, SCL_mav, SC \\ (3) & STD & GSR_mav, SCL_mav, SC \\ (4) & WL & GSR_mav, SCL_mav, SC \\ (5) & SSC & GSR_mav, SCL_mav, SC \\ (6) & WAMP & GSR_mav, SCL_mav, SC \\ (6) & WAMP & GSR_mav, SCL_mav, SC \\ (6) & WAMP & GSR_mav, SCL_mav, SC \\ (7) & NR & GSR_mav \\ (1) & NR & GSR_mav \\ (2) & AVRA & GSR_mav \\ (3) & MAXRA & GSR_mav \\ (3) & MAXRA & GSR_mav \\ (2) & FMN & GSR_mav, SCL_mav, SC \\ (2) & FMN & GSR_mav, SCL_mav, SC \\ (2) & FMN & GSR_mav \\ (2) & FMN & GSR_mav \\ (2) & STD & HR_std \\ (1) & LFI & HR_lfi \end{cases}$	GSR_mav, SCL_mav, SCR_mav
		(1) NR	GSR_nr
	Event related	(2) AVRA	GSR_avra
	Event related (2) AV (3) M.	(3) MAXRA	GSR_maxra
		(1) FMD	GSR_mav, SCL_mav, SCR_mav
	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		GSR_mav, SCL_mav, SCR_mav
		(1) MAV	HR_mav
	Time domain	(1)MAVGSR_mav, SCL_mav, SCR_mar(2)MARGSR_mav, SCL_mav, SCR_mar(3)STDGSR_mav, SCL_mav, SCR_mar(4)WLGSR_mav, SCL_mav, SCR_mar(5)SSCGSR_mav, SCL_mav, SCR_mar(6)WAMPGSR_mav, SCL_mav, SCR_mar(1)NRGSR_nr(2)AVRAGSR_mav, SCL_mav, SCR_mar(3)MAXRAGSR_mav, SCL_mav, SCR_mar(2)FMDGSR_mav, SCL_mav, SCR_mar(2)FMNGSR_mav, SCL_mav, SCR_mar(2)FMNGSR_mav, SCL_mav, SCR_mar(2)STDHR_mav(2)STDHR_std(3)HLRHR_hlr	HR_std
HR			HR_lfi
		HR_hfi	
		(3) HLR	HR_hlr

Table 1. Complete list of the 32 features extracted from the physiological signals.

As shown in Table 2, for each subject, we obtained 15 time windows, with 32 numerical values of the features for each window. Each window has an associated level of anxiety (*ANX*), resulting from the self-assessment. Consequently, there are 105 data points consisting of the feature values and the corresponding self-reported anxiety levels.

Subject Number	Data Window	ANX	Features
	1	010	1, 2, 3,
1	2	010	1, 2, 3,
	•••	••••	
	15	010	
	••••		
7	1 15	0 10	1, 2, 3,

Table 2. Experimental data organization for processing.

A prospective study to identify the features which are useful in estimating anxiety levels began with a Pearson bivariate correlation between anxiety and feature. Table 3 presents a ranking with the features that showed the highest degree of correlation with the subjects' self-expressed anxiety levels.

Feature	Anxiety
SCL_mar	0.715 ***
GSR_mar	0.700 ***
GSR_wamp	0.372 ***
GSR_mav	0.350 ***
SCL_mav	0.345 ***
SCR_wamp	0.341 ***
SCR_fmd	-0.325 **
SCR_fmn	-0.319 **
SCL_wl	0.310 **
GSR_wl	0.309 **
SCL_std	0.305 **
SCR_wl	0.294 **
GSR_std	0.289 **
SCR_std	0.276 **
GSR_maxra	0.257 **
GSR_avra	0.256 **
SCR_mav	0.255 **
HR_lfi	0.212 *
GSR_fmn	-0.202 *
GSR_nr	0.192 *
*** <i>p</i> < 0.001, ** <i>p</i> < 0.0	01, * <i>p</i> < 0.05 (two-tailed).

Table 3. Ranking of features that showed the highest degree of correlation with anxiety.

It can be observed that more than half of the features present a high degree of correlation with the level of anxiety. This suggests the possibility of obtaining an estimate of the anxiety level based on the EDA/HRV features values.

There are two possible approaches for the anxiety estimator design:

- Using a neural network trained with the experimental data;
- Using a mathematical formula identified based on the experimental data.

The first approach is very commonly used today, but requires a much larger volume of training data than is available in this case study, and the computational effort required for the neural network classification can be an issue for the real-time implementation. Therefore, we decided to use the second approach that involves a selection of a set of relevant EDA/HRV features, the proposal of an estimation formula, and the identification of its coefficients.

Our proposed approach involves a regression analysis for the estimation of the relationship between the anxiety intensity (dependent variable) and physiological signal features (independent variables).

The most important step in this analysis is the choice of the regression model. Most regression models are described by the following equation:

$$y = f(x,c) + e, \tag{17a}$$

where *y* represents the dependent variable, *x*—the vector containing independent variables, *c*—the vector of model parameters, and *e* represents an additive error term that may stand in for un-modeled determinants of y or random statistical noise.

In practice, there are specific models known to be suitable for different categories of processes. In our case, the selection of the regression model is difficult to be done because there is no a priori information about the composition of the features set (which are the features that best predict the level of anxiety) and the actual formula of the function f(x, c).

Due to unlimited possibilities to create the model function, we chose to use a second order approximation of the multivariable Taylor series:

$$y = f(a) + (x - a)^T \nabla f(xa) + (x - a)^T H(x_0) (x - a),$$
(17b)

where $\nabla f(a)$ and H(a) are the gradient vector, respectively, Hessian matrix of the estimation function f(x) evaluated in the reference point *a*.

If the reference point is chosen a = 0, due to the symmetry of the Hessian matrix, Equation (17b) can be rewritten as:

$$y = \sum_{\substack{I = 1...P\\J = 1...P\\I \le J}} \alpha_{I,J} x_I x_J + \sum_{I=1}^{r} \beta_I x_I + \gamma,$$
(17c)

Coefficients vector $c = \{ \alpha_{I,J}, \beta_I, \gamma \}$ can be determined using the least squares ethod to minimize the sum of squared residuals:

$$S = \sum_{K=1}^{N} [y(k) - y^{*}(k)]^{2}, \qquad (17d)$$

where y(k) and $y^*(k)$ are the real values of the dependent variable from the available dataset, respectively, the model output values. In our case, because the dependent variable is the anxiety level y(k) = ANX(k)and $y^*(k) = ANX^*(k)$ represents the self-reported, respectively, estimated anxiety levels corresponding to the data point from line *k* in Table 2. Moreover, the regression model function acts as an estimation for anxiety level based on the values of biophysical signals features.

The most difficult part is the selection of the features that compose the estimation function. Using all features would be difficult, due to the high complexity of the estimation function (if all 17 significant features are used, the estimation function would have 171 coefficients).

In order to select the optimal combination of features, we performed an exhaustive search using a MATLAB script specially developed for this purpose. The algorithm used is the following:

- 1. Randomly split the experimental dataset into two subsets: Identification set (IS) containing 90 data points and validation set (VS) containing 15 data points;
- 2. For all combinations of *P* features/from the set of 32 (there are/possible combinations):
 - a. Determine the coefficients of the anxiety estimation function using the current feature set and the information from the IS data points;
 - b. Evaluate the sum of squared residuals;
 - c. Store the current features names, estimation function coefficients/and the sum of squared residuals *S*;
- 3. Find the best selection of features and the corresponding evaluation function coefficients by searching the minimum sum of the squared residuals in the stored data;
- 4. Using the optimal set of features and the corresponding estimation function, evaluate the sum of the squared residuals for the VS data points in order to validate the efficiency of anxiety estimation.

The script was run with a different number of features, starting from P = 1 up to P = 7. For a large number of features, the process of selection became very time-consuming (if P = 8, there are more than 10 million feature combinations to analyze).

A very important part of this algorithm is the splitting of the dataset into the identification set (IS) and validation set (VS). If the identification of these coefficients had been made to minimize the sum of squared residuals for the entire set of experimental data, there would have been a risk of overfitting. Thus, the estimation function would have presented very good performances for the data in this set, but the accuracy of the estimation would vary widely in the case of the data sets generated during new experiments.

The solution used to reduce this phenomenon is to identify the function coefficients using only a part of the data set and validate the estimation accuracy to be done using the rest of the data. The success of this approach depends on how the data set is separated, so that both subsets are representative. If this condition is met, the degree of accuracy obtained for the data in the identification set and those in the validation set is similar.

The attempt to randomly separate the experimental data set obtained in the case study frequently led to excellent results for estimating anxiety on the training set, but low accuracy for the validation set (an example is shown in Figure 8).



Figure 8. Estimation errors distribution for an improper data set split and 7 features (**a**) for identification subset; (**b**) for validation subset.

The occurrence frequency of these situations, as well as the level of accuracy difference between the estimates on the two subsets of data, increases with the increase in the number of parameters used and consequently in the number of coefficients in the estimation function. One explanation for this phenomenon would be that a more complex function is much more likely to overfit to a small set of identification data. For this reason, the use of a larger number of parameters is not indicated if a large number of experimental data is not available.

To reduce the risk of choosing unrepresentative data subsets, we introduced a condition for their generation, namely, a distribution of reported anxiety levels relatively identical at the level of the two data subsets.

The results presented in Figure 8 correspond to the following distribution of anxiety levels in the two subsets (Table 4).

Reported Anxiety Level	Occurrences in the Whole Data Set <i>NT_K</i>	Occurrences in the Identification Subset NI_K =round $\left(\frac{90}{105}NT_K\right)$	Occurrences in the Validation Subset NV_K =round $\left(\frac{15}{105}NT_K\right)$
0	41	35	6
1	16	14	2
2	12	10	2
3	9	8	1
4	7	6	1
5	4	3	1
6	4	3	1
7	2	2	0
8	5	4	1
9	2	2	0
10	3	3	0

Table 4. Distribution of the reported anxiety levels

8. Results

In the process of finding the best combination of features to be used in the anxiety estimation function, we performed an exhaustive search over all possible combinations of features. The number of features taken into account as input variables into the estimation function was initially selected to 1 (resulting $C_{32}^1 = 32$ possible choices) and gradually increased up to 7 (resulting $C_{32}^7 = 3365856$ possible combination choices). For each number of input variables, the optimal combination of features and corresponding estimation function coefficients were selected based on the minimization of the sum of squared differences between the reported and estimated anxiety level.

In order to obtain reliable results, the experimental data set was randomly divided into two subsets. The first subset (IS), containing 90 data points, was used for the identification of estimating function coefficients. The second subset (VS), containing 15 data points, was used to assess the accuracy of the anxiety estimation on a different data set than the one used to determine the estimation function.

The accuracy of the estimation was also assessed by means of the sum of the squared differences between the reported and estimated anxiety level. This criterion was evaluated separately for the data in the identification set (IS), validation set (VS), as well as for the entire data set (IS + VS).

Table 5 presents the best accuracy obtained by the optimal estimation function for a different number of features. Moreover, this table presents the names of the features that give that accuracy level.

Number of		Sum of	Squared Estimat	ion Errors
Parameters Used	Parameter	Whole Dataset	Identification Dataset	Validation Dataset
1	SCL_mar	3.827	3.782	4.092
2	HR_mav, GSR_mar	3.059	2.921	3.884
3	HR_mav, GSR_std, GSR_mar	2.663	2.519	3.526
4	HR_mav,GSR_mar, SCR_mar, SCR_fmd	2.041	2.050	1.983
5	HR_mav, SCR_mav, SCL_mar, GSR_wamp, SCR_wamp	1.656	1.674	1.546
6	HR_mav, HR_std, SCR_mav, SCL_mar, GSR_wamp, SCR_wamp	1.286	1.327	1.044
7	HR_mav, HR_std, SCR_mav, SCL_mar, SCR_mar, GSR_wamp, SCR_wamp	1.031	1.046	0.940

Table 5. Estimation performance for different number of selected features

By analyzing the data in the table, we can observe the small difference between the values evaluated for the set of identification data and the validation one. This observation confirms that each identified function is able to provide estimates with a similar degree of accuracy, not only for the measurements performed in this case, but also in the case of other measurements that will be performed during future VR experiments.

Moreover, as expected, the accuracy of the estimate increases with the increasing number of parameters used. Because evaluating the estimation function does not involve a high computational effort, the function with the highest number of parameters can be used for maximum estimation accuracy.

Another performance indicator is the accuracy of anxiety classification. It can be evaluated by the confusion matrix and the overall classification accuracy (percentage of correct classified anxiety levels).

In our study, we tested anxiety classifications, levels 2 and 3. The intensity intervals definition of the classes are:

- [0...5] for "low" and (5... 10] for "high"
- [0...3] for "low", (3...7] for "mild" and (7...10] for "high".

The results of estimation evaluation for different numbers of features selected as input values are presented in Tables 6–12. Each table presents the histogram associated with the distribution of estimation errors (on the scale of 0 to10), confusion matrix for levels 2 and 3, and corresponding overall classification accuracy.

30	Mean: -0.01]					Low	0.85	0.15	0.00
20 Intences	Deviation: 1.97	eal	Low	0.96	0.04	Real	Mild	0.59	0.23	0.18
r of occi		Ř	High	0.37	0.63		High	0.00	0.70	0.30
Indmb Indmb				Low	High	-		Low	Mild	High
~				Estimated					Estimated	
U	-6 -5 -4 -3 -2 -1 0 1 2 3 4 5 6 Error values			Accuracy:	90.48%			I	Accuracy: (69.52%

Table 6. Distribution of estimation errors and confusion matrices for levels 2 and 3—using one feature.

25		T					Low	0.95	0.05	0.00
20 Lences	Mean: 0.07 Deviation: 1.76	eal	Low	0.98	0.02	Real	Mild	0.29	0.65	0.06
15 Jo		Å	High	0.44	0.56	_	High	0.00	0.50	0.50
umber 2				Low	High	-		Low	Mild	High
z				Estimated					Estimated	
0	-6 -5 -4 -3 -2 -1 0 1 2 3 4 5 6			Accuracy: 9	91.43%			A	Accuracy: 8	85.71%
	Error values			5					5	

Table 7. Distribution of estimation errors and confusion matrices for levels 2 and 3—using two features.

Table 8. Distribution of estimation errors and confusion matrices for levels 2 and 3—using three features.

25 Mean: 0.10	ור					Low	0.96	0.04	0.00
20 Deviation: 1.64	eal	Low	0.99	0.01	Real	Mild	0.41	0.47	0.12
0 10 -	R	High	0.25	0.75	_	High	0.00	0.40	0.60
	_		Low	High	-		Low	Mild	High
			Estimated					Estimated	
-6 -5 -4 -3 -2 -1 0 1 2 3 4 5 6 Error values			Accuracy:	95.24%			I	Accuracy:	84.76%

Table 9. Distribution of estimation errors and confusion matrices for levels 2 and 3—using four features.



Table 10. Distribution of estimation errors and confusion matrices for levels 2 and 3—using five features.







30 0	Mean: 0.02					_	Low	0.96	0.04	0.00
nrencei	Deviation: 1.02	Lo Lo	W	0.99	0.01	Real	Mild	0.23	0.65	0.12
r of occ		بع Hi٤	gh	0.06	0.94	-	High	0.00	0.10	0.90
equny				Low	High			Low	Mild	High
~ 0				Estimated					Estimated	
0	-6 -5 -4 -3 -2 -1 0 1 2 3 4 5 6			Accuracy: 9	8.10%			1	Accuracy: 9	90.48%

Table 12. Distribution of estimation errors and confusion matrices for levels 2 and 3—using seven features.

9. Discussion

The protocol proposed in this article was developed following the data analysis of the experiments performed in order to optimize the accuracy of anxiety level estimation during VR based exposure therapy. The elements of this protocol were designed in order to allow estimations in real-time with low latencies. Therefore, although the proposed protocol largely follows the structure of most protocols present in the literature (data acquisition, feature extraction, estimation/classification of anxiety level), the need to make real-time estimates led to the use of a shorter time window for feature extraction and a regression model for estimating the level of anxiety (as an alternative to the computationally intensive methods currently used for classification).

The performance evaluation of the proposed anxiety estimation protocol can be made by analyzing two categories of parameters: Estimation error characteristics and classification accuracy.

Analyzing the estimation error, the standard deviation value of 1.02 obtained in the case of an estimation function with seven features makes us to expect that 95% of the estimates do not differ by more than two units from the real values that can be considered a good performance.

If only two classes are used, the classification accuracy evaluated by the confusion matrices is very good even for regression models that use a low number of features (with only three features, the "low" anxiety level was correctly classified in 99% of cases and the "high" level in 75%).

Classification in three classes proves to be a more difficult task, a good accuracy level being reached for a regression model that uses six features (95% correct classification for "low" anxiety level, 76% for "mild" and 100% for "high). Moreover, it can be observed that most of the misclassified values belong to the "mild" class.

Although increasing the number of features has the effect of reducing the estimation error, it can be observed that this improvement does not always reflect in classification accuracy. For example, increasing the number of features from 6 to 7 reduces the standard deviation of the estimation errors from 1.14 to 1.02, but the overall classification accuracy decrease from 92.38% to 90.48%. This phenomenon can be explained by small errors in estimating the level of anxiety can lead to incorrect classifications if its value is located around the threshold.

In the case mentioned above, when seven features were used for estimation, 3 of the 17 levels of anxiety in the "mild" class were incorrectly classified as "low" or "high" as follows: One anxiety level was misclassified as "low" because its estimated value was 2.91 (slight below the threshold value of 3) and two anxiety levels were misclassified as "high" because their estimated values were 7.12, respectively, 7.37 (slight above the threshold value of 7).

Another important observation related to the classification performance is the high level of the overall accuracy obtained even if the correct classification rate is relatively low, especially in the case of "mild" anxiety levels. The cause of this overestimation of accuracy is the unbalanced distribution of anxiety levels in the data set. The prevalence of "low" anxiety levels is very high (78 out of 105) and the accuracy of classifications in these cases is very good (around 95%) that reduces the impact of the few misclassified levels from the other classes.

9.1. Comparison with Other Studies Results

Anxiety disorders affect about one-third of the population during their lifetime. For this reason, researchers make efforts in order to identify all possible causes, improve diagnostic accuracy, and find more efficient treatment protocols. Because evaluating the anxiety level is very important, especially during treatment sessions, multiple studies have been done in this area. Most of the recent studies are focused on the use of various physiological signals in order to evaluate anxiety intensity.

Therefore, an objective evaluation of the results obtained in our study requires a comparison with several works. We selected articles that present different anxiety classification methods tested with both real and VR-induced stimuli. Table 13 presents a summary of the comparison between different classifiers found in the literature and our results.

The first observation related to this comparison is that almost all studies use machine learning approaches for classifier implementation. Our approach is different because we chose to estimate anxiety level based on a regression model and after that to decide an affiliation to different classes by simple comparisons with threshold values.

Another observation is related to the length of the time window used for feature extraction and classification. In most of the studies, the time window is relatively large (over two min). A longer time window increases evaluation precision for the frequency-related features, especially if the frequency ranges involved are in the sub-hertz domain. Moreover, large time windows can reduce the effect of the artifacts that may occur during signal acquisition.

In our study, we aimed to perform the anxiety level estimation in real-time because this estimation will be used as feedback to adapt the VR exposure scenario during treatment sessions. In order to obtain a fast response of the estimates to the changes in patient anxiety state, it is necessary to use time windows as small as possible, which led us to choose a time window of 5 s.

Comparative evaluation of the performances obtained in our study based only on the overall accuracy of classification is not very relevant. The overestimation of classification accuracy (94.29% for two classes and 92.38% for three classes) due to the unbalanced distribution of the anxiety levels in our data set requires a more detailed comparison of performance based on confusion matrices.

The best accuracy level was obtained with a regression model that have six features as input variables. According to confusion matrices, the lowest rates of correct classification are:

- 81% for the "high" class—for level 2 classification;
- 76% for the "mild" class—for level 3 classification.

It can be observed that these values are close to the average accuracy levels obtained in the studies presented for comparison in Table 13, therefore we can conclude that our approach is suitable for anxiety level detection and classification.

9.2. Limitations

The most important limitation in our study came from the reduced number of subjects involved in the case study. Moreover, only two subjects experienced a severe form of height intolerance that led to the unbalanced distribution of the anxiety levels in our dataset. Therefore, the inclusion of more subjects with different forms of height intolerance is required in order to obtain more reliable results.

This study did not address the effects of possible artifacts induced by the movements of the subjects. The most important source of movement artifacts is the handling of the VR device controllers that can affect the quality of GSR electrodes contact with skin.

Furthermore, in the case of the GSR signal, the extraction of the physiological parameters involved the decomposition and analysis of the signal based on the Continuous Decomposition Analysis (CDA). This algorithm has a high degree of complexity as well a non-causal character (the determination of the current samples of tonic and phasic components requires knowing several future samples from the analyzed EDA signal) that limits its use in real-time anxiety classification.

Study	Application	Physiological Signals	Method	No. of Subjects	No. of Classes	Time Window	Accuracy
	Public speaking				2 (high anxiety/calmness)		86.70%
Zhang et.al. [102]	anxiety	GSR	Neural Network	22	2 (high anxiety/ low anxiety	300 s	78.83%
Šalkevicius et al. [103]	Public speaking anxiety (VR exposure)	GSR, Blood Volume Pulse (BVP), skin temperature	SVM classifier	30	4 (low, mild, moderate, high)	20 s	86.30%
Sandulescu et al. [104]	Trier Social Stress Test (TSST)	GSR, BVP	SVM	5	2 (stressed, not stressed)	300 s (stress)/ 120 s (no stress)	73.26–83.08% (subject dependent)
	Arachno-phobia	GSR, HR, HRV, respiration	Bagged Trees			10 s	89.80%
					2 (low, high)	60 s	90.90%
Ihmig et al. [105]	(VR exposure)			80	3 (low, medium,	10 s	74.40%
					high)	60 s	73.40%
		ECG. GSR.	SVM		3 (low, medium,		89.70%
Chen et al. [106]	Driving stress	respiration	ELM	9	high)	100 s	89.20%
Liu et.al. [107]	Driving stress	GSR	Linear discriminant analysis	11	3 (low, medium, high)	300 s	81.82%
Sharma et al. [108]	Social Anxiety	GSR	MLP	31	2 (low, high)	120 s	85.70%
	Fear of heights		Regression model.		2 (low high)		94.29%
Our study	(VR exposure)	GSR, HRV	threshold comparison	7	3 (low, mild, high)	5 s	92.38%

 Table 13. Comparison between different classifiers found in literature and our results.

In this context, we notice the need to develop an algorithm (possibly suboptimal) for the decomposition and analysis of the EDA signal in real-time. The task of developing this algorithm is simplified by the procedure for selecting the optimal parameters for estimating the level of anxiety indicated the need to use only parameters derived from the decomposition into tonic and phasic components and not those derived from events identification.

10. Conclusions

Virtual reality is a viable alternative to in-vivo exposure for treating phobias. The possibility of adapting the scenario during exposure therapy, offered by virtual reality, is an important element for increasing the effectiveness of the therapy. The scenario's adaptation involves modulating the intensity of the phobic stimuli according to the level of anxiety induced by them. For this reason, an objective and real-time assessment of anxiety level based on the analysis of measurable physiological parameters is an essential condition for implementing this type of therapy.

In this article, we have presented a protocol for evaluating the level of anxiety, to be used in phobia therapy with exposure in virtual reality. This protocol contains recommendations on the preparation and measurement of physiological parameters during the therapy session and proposes a method of processing the acquired data in order to estimate the level of anxiety.

Both the recommendations and the estimation method were elaborated following the analysis of the experimental data obtained during the exposure of seven subjects to stimuli that trigger fear of heights.

The method for estimating the level of anxiety is based on the real-time evaluation of a nonlinear function having as parameters a set of features extracted from the physiological signals measured during the therapy session.

The larger volume of data obtained from a higher number of subjects will allow us to increase the accuracy of the estimates and will provide a higher degree of certainty for protocol validation. Moreover, a larger volume of data will allow us to train a neural network as an alternative to estimate the level of anxiety, and then perform a comparative analysis of the results in terms of accuracy and implementation complexity.

The integration of protocol at biophysical measurements in Virtual Reality environments for anxiety detection can have numerous implications in both practice and research.

For clinical applicability, we can mention the treatment of anxiety by the VR immersion therapy, ideal being the development of a system with the dual operation: (i) For subclinical forms, the system can be implemented autonomously, like a biofeedback application or possibly with remote targeting by the therapist; (ii) for the clinical forms, the direct intervention of the therapist is needed, with a control panel for the permanent monitoring of the evolution of the treatment and of the patient's condition (by monitoring the physiological parameters).

As a research target, the protocol can be used as a method of developing and testing applications that require real-time evaluation of the user's emotional response (medical, educational applications, military, or sports training).

Author Contributions: Conceptualization, L.P., O.M., G.M.; methodology, L.P., C.P., O.M.; software, L.P., C.P.; validation, O.M., G.M., A.M., F.M., M.L.; formal analysis, L.P.; investigation, L.P., C.P., O.M., G.M.; resources, A.M., F.M., M.L.; data curation, L.P., C.P.; writing—original draft preparation, L.P., C.P., O.M., G.M.; writing—review and editing, O.M.; visualization, L.P., C.P.; supervision, A.M., F.M.; project administration, A.M.; funding acquisition, M.L. All authors have read and agreed to the published version of the manuscript.

Funding: This work was supported by a grant of the Romanian Ministery of Research and Innovation, CCCDI—UEFISCDI, project number 43PTE/2020, "PhoVR—Tratarea Imersiva a Fobiilor prin Realitate Virtuala Adaptiva si Biofeedback"/"PhoVR—Immersive Treatment of Phobias through Adaptive Virtual Reality and Biofeedback", within PNCDI III. The work has been funded by UEFISCDI project 1/2018 and UPB CRC Research Grant 2017.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Fairclough, S.H. Physiological computing: Interfacing with the human nervous system. In *Sensing Emotions;* Springer: Dordrecht, The Netherlands, 2010; pp. 1–20.
- Munoz, J.E.; Paulino, T.; Vasanth, H.; Baras, K. PhysioVR: A novel mobile virtual reality framework for physiological computing. In Proceedings of the 2016 IEEE 18th International Conference on E-Health Networking, Applications and Services (Healthcom), Munich, Germany, 14–16 September 2016.
- 3. Wallbott, H.G.; Scherer, K.R. Assessing emotion by questionnaire. In *The Measurement of Emotions*; Elsevier: Amsterdam, The Netherlands, 1989; pp. 55–82.
- 4. Dzedzickis, A.; Kaklauskas, A.; Bucinskas, V. Human emotion recognition: Review of sensors and methods. *Sensors* **2020**, *20*, 592. [CrossRef] [PubMed]
- 5. Bradley, M.M.; Codispoti, M.; Cuthbert, B.N.; Lang, P.J. Emotion and motivation I: Defensive and appetitive reactions in picture processing. *Emotion* **2001**, *1*, 276–298. [CrossRef] [PubMed]
- 6. Boucsein, W. Principles of electrodermal phenomena. In *Electrodermal Activity*; Springer: Boston, MA, USA, 2012; pp. 1–86.
- 7. Ekman, P. An argument for basic emotions. *Cogn. Emot.* **1992**, *6*, 169–200. [CrossRef]
- 8. Plutchik, R. The Nature of Emotions. Am. Sci. 2001, 89, 344–350. [CrossRef]
- 9. Izard, C.E. Emotion theory and research: Highlights, unanswered questions, and emerging issues. *Annu. Rev. Psychol.* **2008**, *60*, 1–25. [CrossRef]
- 10. Lang, P.J. The emotion probe: Studies of motivation and attention. Am. Psychol. 1995, 50, 372–385. [CrossRef]
- 11. Shu, L.; Xie, J.; Yang, M.; Li, Z.; Li, Z.; Liao, D.; Xu, X.; Yang, X. A Review of Emotion Recognition Using Physiological Signals. *Sensors* **2018**, *18*, 2074. [CrossRef]
- 12. Mehrabian, A. Comparison of the PAD and PANAS as models for describing emotions and for differentiating anxiety from depression. *J. Psychopathol. Behav. Assess.* **1997**, *19*, 331–357. [CrossRef]
- 13. Steimer, T. The biology of fear- and anxiety-related behaviors. Dialogues Clin. Neurosci. 2002, 4, 231-249.
- 14. Adolphs, R. The biology of fear. Curr. Biol. 2013, 23, 79–93. [CrossRef]
- Lisetti, C.L. Affective computing: By Rosalind Picard. Cambridge, Mass: MIT Press. *Pattern Anal. Appl.* 1998, 1, 71–73. [CrossRef]
- 16. Lang, P.J. *International Affective Picture System (IAPS): Technical Manual and Affective Ratings*; The Center for Research in Psychophysiology, University of Florida: Gainesville, FL, USA, 1995.
- 17. Bai, L.; Ma, H.; Huang, H.; Luo, Y. The compilation of Chinese emotional picture system—A trial of 46 Chinese college students. *Chin. J. Ment. Health* **2005**, *19*, 719–722.
- 18. Bradley, M.M.; Lang, P.J. *The International Affective Digitized Sounds (2nd Edition; IADS-2): Affective Ratings of Sounds and Instruction Manual;* Technical Report B-3; University of Florida: Gainesville, FL, USA, 2007.
- 19. The Center for the Study of Emotion and Attention. Available online: https://csea.phhp.ufl.edu/Media.html (accessed on 20 September 2020).
- 20. Bradley, M.M.; Lang, P.J. *Affective Norms for English Words (ANEW): Instruction Manual and Affective Ratings;* Technical Report C-3; UF Center for the Study of Emotion and Attention: Gainesville, FL, USA, 2017.
- 21. Bradley, M.M.; Lang, P.J. Affective Norms for English Text (ANET): Affective Ratings of Text and Instruction Manual; Tech. Rep. No. D-1; University of Florida: Gainesville, FL, USA, 2007.
- 22. Nasoz, F.; Alvarez, K. Emotion Recognition from Physiological Signals for Presence Technologies. *Int. J. Cogn. Technol. Work* 2003, *6*, 4–14. [CrossRef]
- 23. Koelstra, S.; Muhl, C.; Soleymani, M.; Lee, J.S.; Yazdani, A.; Ebrahimi, T.; Pun, T.; Nijholt, A.; Patras, I. DEAP: A Database for Emotion Analysis; Using Physiological Signals. *IEEE Trans. Affect. Comput.* **2012**, *3*, 18–31. [CrossRef]
- 24. Herbelin, B.; Benzaki, P.; Riquier, F.; Renault, O.; Grillon, H.; Thalmann, D. Using physiological measures for emotional assessment: A computer-aided tool for cognitive and behavioural therapy. *Int. J. Disabil. Hum. Dev.* **2005**, *4*. [CrossRef]
- 25. Bradley, M.M.; Lang, P.J. Measuring emotion: The self-assessment manikin and the semantic differential. *J. Behav. Ther. Exp. Psychiatry* **1994**, 25, 49–59. [CrossRef]
- 26. Betella, A.; Verschure, P. The affective slider: A digital self-assessment scale for the measurement of human emotions. *PLoS ONE* **2016**, *11*, e0148037. [CrossRef]

- 27. Sachs, G.; Virtual and Augmented Reality. Understanding the Race for the Next Computing Platform. 2016. Available online: https://trends.ifla.org/node/436 (accessed on 1 December 2020).
- 28. Draper, J.V.; Kaber, D.B.; Usher, J.M. Telepresence. Hum. Factors 1998, 40, 354–375. [CrossRef]
- 29. Ravaja, N.; Salminen, M.; Holopainen, J.; Saari, T.; Laarni, J.; Jarvinen, A. Emotional response patterns and sense of presence during video games: Potential criterion variables for game design. In Proceedings of the Third Nordic Conference on Human-Computer Interaction, Tampere, Finland, 23–27 October 2004; pp. 339–347.
- 30. Felnhofer, A.; Kothgassner, O.D.; Schmidt, M.; Heinzle, A.-K.; Beutl, L.; Hlavacs, H.; Kryspin-Exner, I. Is virtual reality emotionally arousing? Investigating five emotion inducing virtual park scenarios. *Int. J. Hum. Comput. Stud.* **2015**, *82*, 48–56. [CrossRef]
- 31. Wiederhold, B.K.; Jang, D.P.; Kim, S.I.; Wiederhold, M.D. Physiological monitoring as an objective tool in virtual reality therapy. *Cyberpsychol. Behav.* **2002**, *5*, 77–82. [CrossRef]
- 32. Riva, G.; Mantovani, F.; Capideville, C.S.; Preziosa, A.; Morganti, F.; Villani, D.; Gaggioli, A.; Botella, C.; Alcañiz, M. Affective interactions using virtual reality: The link between presence and emotions. *Cyberpsychol. Behav.* **2007**, *10*, 45–56. [CrossRef] [PubMed]
- Jang, D.P.; Kim, I.Y.; Nam, S.W.; Wiederhold, B.K.; Wiederhold, M.D.; Kim, S.I. Analysis of physiological response to two virtual environments: Driving and flying simulation. *Cyberpsychol. Behav.* 2002, 5, 11–18. [CrossRef] [PubMed]
- 34. Meehan, M.; Insko, B.; Whitton, M.; Brooks, F.P., Jr. Physiological measures of presence in stressful virtual environments. *ACM Trans. Graph.* **2002**, *21*, 645–652. [CrossRef]
- 35. Peterson, S.M.; Furuichi, E.; Ferris, D.P. Effects of virtual reality high heights exposure during beam-walking on physiological stress and cognitive loading. *PLoS ONE* **2018**, *13*, e0200306. [CrossRef]
- 36. Fabri, M.; Moore, D.; Hobbs, D. Mediating the expression of emotion in educational collaborative virtual environments: An experimental study. *Virtual Real.* **2004**, *7*, 66–81. [CrossRef]
- Fabri, M.; Moore, D.J.; Hobbs, D.J. The emotional avatar: Non-verbal communication between inhabitants of collaborative virtual environments. In *Gesture-Based Communication in Human-Computer Interaction*; Springer: Berlin/Heidelberg, Germany, 1999; pp. 269–273.
- 38. Moser, E.; Derntl, B.; Robinson, S.; Fink, B.; Gur, R.C.; Grammer, K. Amygdala activation at 3T in response to human and avatar facial expressions of emotions. *J. Neurosci. Methods* **2007**, *161*, 126–133. [CrossRef]
- 39. Roy, M.J.; Costanzo, M.E.; Jovanovic, T.; Leaman, S.; Taylor, P.; Norrholm, S.D.; Rizzo, A.A. Heart rate response to fear conditioning and virtual reality in subthreshold PTSD. *Stud. Health Technol. Inform.* **2013**, *191*, 115–119.
- Rizzo, A.; Morie, J.; Williams, J.; Pair, J.; Buckwalter, J. Human Emotional State and its Relevance for Military VR Training. In Proceedings of the 11th International Conference on Human Computer Interaction, Las Vegas, NV, USA, 22–27 July 2005; pp. 1–10.
- Zhang, W.; Shu, L.; Xu, X.; Liao, D. Affective Virtual Reality System (AVRS): Design and Ratings of Affective VR Scences. In Proceedings of the International Conference on Virtua lReality and Visualization, Zhengzhou, China, 21–22 October 2017.
- 42. Appypie. Available online: https://www.appypie.com/ (accessed on 20 September 2020).
- 43. Sequeira, H.; Hot, P.; Silvert, L.; Delplanque, S. Electrical autonomic correlates of emotion. *Int. J. Psychophysiol.* **2009**, *71*, 50–56. [CrossRef]
- 44. Baker, L.B. Physiology of sweat gland function: The roles of sweating and sweat composition in human health. *Temperature* **2019**, *6*, 211–259. [CrossRef]
- 45. Harker, M. Psychological sweating: A systematic review focused on aetiology and cutaneous response. *Ski. Pharmacol. Physiol.* **2013**, *26*, 92–100. [CrossRef]
- 46. Posada-Quintero, H.F.; Chon, K.H. Innovations in electrodermal activity data collection and signal processing: A systematic review. *Sensors* **2020**, *20*, 479. [CrossRef] [PubMed]
- 47. Sarchiapone, M.; Gramaglia, C.; Iosue, M.; Carli, V.; Mandelli, L.; Serretti, A.; Marangon, D.; Zeppegno, P. The association between electrodermal activity (EDA), depression and suicidal behaviour: A systematic review and narrative synthesis. *BMC Psychiatry* **2018**, *18*, 22. [CrossRef] [PubMed]
- 48. Edelberg, R. Local electrical response of the skin to deformation. *J. Appl. Physiol.* **1973**, *34*, 334–340. [CrossRef] [PubMed]

- Heiervang, E.; Stevenson, J.; Lund, A.; Hugdahl, K. Behaviour problems in children with dyslexia. Nord. J. Psychiatry 2001, 55, 251–256. [CrossRef] [PubMed]
- 50. Leiner, D.; Fahr, A.; Früh, H. EDA positive change: A simple algorithm for electrodermal activity to measure general audience arousal during media exposure. *Commun. Methods Meas.* **2012**, *6*, 237–250. [CrossRef]
- Boucsein, W.; Fowles, D.C.; Grimnes, S.; Ben-Shakhar, G.; Roth, W.T.; Dawson, M.E.; Filion, D.L. Society for Psychophysiological Research Ad Hoc Committee on Electrodermal Measures. Publication recommendations for electrodermal measurements: Publication standards for EDA. *Psychophysiology* 2012, 49, 1017–1034.
- Braithwaite, J.; Watson, D.; Robert, J.; Mickey, R. A Guide for Analysing Electrodermal Activity (EDA) & Skin Conductance Responses (SCRs) for Psychological Experiments; University of Birmingham: Birmingham, AL, USA, 2013.
- 53. Harmon-Jones, E.; Beer, J.S. Methods in Social Neuroscience; The Guilford Press: New York, NY, USA, 2009.
- 54. Crider, A. Personality and electrodermal response lability: An interpretation. *Appl. Psychophysiol. Biofeedback* **2008**, *33*, 141–148. [CrossRef]
- Crider, A.; Kremen, W.S.; Xian, H.; Jacobson, K.C.; Waterman, B.; Eisen, S.A.; Tsuang, M.T.; Lyons, M.J. Stability, consistency, and heritability of electrodermal response lability in middle-aged male twins. *Psychophysiology* 2004, *41*, 501–509. [CrossRef]
- Thorell, L.H.; Wolfersdorf, M.; Straub, R.; Steyer, J.; Hodgkinson, S.; Kaschka, W.P.; Jandl, M. Electrodermal hyporeactivity as a trait marker for suicidal propensity in uni- and bipolar depression. *J. Psychiatr. Res.* 2013, 47, 1925–1931. [CrossRef]
- 57. Wendt, J.; Lotze, M.; Weike, A.I.; Hosten, N.; Hamm, A.O. Brain activation and defensive response mobilization during sustained exposure to phobia-related and other affective pictures in spider phobia. *Psychophysiology* **2008**, *45*, 205–215. [CrossRef]
- Wang, H.; Prendinger, H.; Igarashi, T. Communicating emotions in online chat using physiological sensors and animated text. In Proceedings of the 2004 Conference on Human Factors and Computing Systems—CHI'04, Vienna, Austria, 24–29 April 2004.
- 59. Dawson, M.E.; Schell, A.M.; Filion, D.L. The Electrodermal System. In *Handbook of Psychophysiology*; Cambridge University Press: Cambridge, UK, 2016; pp. 217–243.
- 60. Boucsein, W. Applications of Electrodermal Recording. In *Electrodermal Activity*; Springer: Boston, MA, USA, 2012; pp. 259–523.
- 61. Greco, A.; Valenza, G.; Lanata, A.; Scilingo, E.P.; Citi, L. CvxEDA: A convex optimization approach to electrodermal activity processing. *IEEE Trans. Bio-Med. Eng.* **2016**, *63*, 797–804. [CrossRef] [PubMed]
- 62. Baevsky, R.M. Institute of Biomedical Problems of the Russian Academy of Sciences, & Chernikova, A.G. Heart rate variability analysis: Physiological foundations and main methods. *Cardiometry* **2017**, *10*, 66–76.
- 63. Cohen, M.A.; Taylor, J.A. Short-term cardiovascular oscillations in man: Measuring and modelling the physiologies. *J. Physiol.* **2002**, *542*, 669–683. [CrossRef] [PubMed]
- Porges, S.W.; Byrne, E.A. Research methods for measurement of heart rate and respiration. *Biol. Psychol.* 1992, 34, 93–130. [CrossRef]
- 65. Porges, S.W. The polyvagal perspective. Biol. Psychol. 2007, 74, 116–143. [CrossRef] [PubMed]
- 66. Thayer, J.F.; Lane, R.D. A model of neurovisceral integration in emotion regulation and dysregulation. *J. Affect. Disord.* **2000**, *61*, 201–216. [CrossRef]
- 67. Kemp, A.H.; Koenig, J.; Thayer, J.F. From psychological moments to mortality: A multidisciplinary synthesis on heart rate variability spanning the continuum of time. *Neurosci. Biobehav. Rev.* 2017, *83*, 547–567. [CrossRef]
- 68. Haiblum-Itskovitch, S.; Czamanski-Cohen, J.; Galili, G. Emotional response and changes in heart rate variability following art-making with three different art materials. *Front. Psychol.* **2018**, *9*. [CrossRef]
- 69. Chalmers, J.A.; Quintana, D.S.; Abbott, M.J.-A.; Kemp, A.H. Anxiety disorders are associated with reduced heart rate variability: A meta-analysis. *Front. Psychol.* **2014**, *5*, 80. [CrossRef]
- Meerwijk, E.L.; Chesla, C.A.; Weiss, S.J. Psychological pain and reduced resting-state heart rate variability in adults with a history of depression: Psychological pain and resting-state physiology. *Psychophysiology* 2014, 51, 247–256. [CrossRef]
- 71. Williams, D.P.; Cash, C.; Rankin, C.; Bernardi, A.; Koenig, J.; Thayer, J.F. Resting heart rate variability predicts self-reported difficulties in emotion regulation: A focus on different facets of emotion regulation. *Front. Psychol.* **2015**, *6*, 261. [CrossRef] [PubMed]

- 72. Yaroslavsky, I.; Bylsma, L.M.; Rottenberg, J.; Kovacs, M. Combinations of resting RSA and RSA reactivity impact maladaptive mood repair and depression symptoms. *Biol. Psychol.* **2013**, *94*, 272–281. [CrossRef] [PubMed]
- 73. Adrian, M.; Zeman, J.; Veits, G. Methodological implications of the affect revolution: A 35-year review of emotion regulation assessment in children. *J. Exp. Child Psychol.* **2011**, *110*, 171–197. [CrossRef] [PubMed]
- Marzbani, H.; Marateb, H.R.; Mansourian, M. Methodological Note: Neurofeedback: A Comprehensive Review on System Design, Methodology and Clinical Applications. *Basic Clin. Neurosci.* 2016, 7, 143–158. [CrossRef] [PubMed]
- 75. Ferche, O.; Moldoveanu, A.; Cintezã, D.; Toader, C.; Moldoveanu, F.; Voinea, A.; Taslitchi, C. From neuromotor command to feedback: A survey of techniques for rehabilitation through altered perception. In Proceedings of the 2015 E-Health and Bioengineering Conference (EHB), Iasi, Romania, 19–21 November 2015; pp. 1–4.
- 76. Moldoveanu, A.; Ferche, O.; Moldoveanu, F.; Lupu, R.; Cinteză, D.; Irimia, D.C.; Toader, C. The TRAVEE System for a Multimodal Neuromotor Rehabilitation. *IEEE Access* **2015**, *7*, 8151–8171. [CrossRef]
- 77. Urry, H.L.; Nitschke, J.B.; Dolski, I.; Jackson, D.C.; Dalton, K.M.; Mueller, C.J.; Rosenkranz, M.A.; Ryff, C.D.; Singer, B.H.; Davidson, R.J. Making a life worth living: Neural correlates of well-being: Neural correlates of well-being. *Psychol. Sci.* 2004, 15, 367–372. [CrossRef]
- Tomarken, A.J.; Davidson, R.J.; Henriques, J.B. Resting frontal brain asymmetry predicts affective responses to films. *J. Personal. Soc. Psychol.* 1990, 59, 791–801. [CrossRef]
- 79. Mauss, I.B.; Robinson, M.D. Measures of emotion: A review. Cogn. Emot. 2009, 23, 209–237. [CrossRef]
- 80. Kim, G.J. Human–Computer Interaction: Fundamentals and Practice; CRC Press: Boca Raton, FL, USA, 2015.
- 81. Vourvopoulos, A.; Faria, A.L.; Cameirao, M.S.; Bermudez i Badia, S. RehabNet: A distributed architecture for motor and cognitive neuro-rehabilitation. In Proceedings of the 2013 IEEE 15th International Conference on E-Health Networking, Applications and Services (Healthcom 2013), Lisbon, Portugal, 9–12 October 2013.
- 82. Li, B.J.; Bailenson, J.N.; Pines, A.; Greenleaf, W.J.; Williams, L.M. A public database of immersive VR videos with corresponding ratings of arousal, valence, and correlations between head movements and self report measures. *Front. Psychol.* **2017**, *8*. [CrossRef]
- Catai, A.M.; Pastre, C.M.; de Godoy, M.F.; da Silva, E.; de Medeiros Takahashi, A.C.; Vanderlei, L.C.M. Heart rate variability: Are you using it properly? Standardisation checklist of procedures. *Braz. J. Phys. Ther.* 2020, 24, 91–102. [CrossRef]
- 84. Sharma, K.; Castellini, C.; van den Broek, E.L.; Albu-Schaeffer, A.; Schwenker, F. A dataset of continuous affect annotations and physiological signals for emotion analysis. *Sci. Data* **2019**, *6*, 196. [CrossRef] [PubMed]
- 85. IMotions. Galvanic Skin Response—The Complete Pocket Guide. 2020. Available online: https://imotions. com/blog/galvanic-skin-response/ (accessed on 1 December 2020).
- 86. Posada-Quintero, H.F.; Florian, J.P.; Orjuela-Cañón, A.D.; Chon, K.H. Electrodermal activity is sensitive to cognitive stress under water. *Front. Physiol.* **2017**, *8*, 1128. [CrossRef] [PubMed]
- 87. Moghimi, M.; Stone, R.; Rotshtein, P. Affective Recognition in Dynamic and Interactive Virtual Environments. *IEEE Trans. Affect. Comput.* **2020**, *11*, 45–62. [CrossRef]
- Benedek, M.; Kaernbach, C. A continuous measure of phasic electrodermal activity. J. Neurosci. Methods 2010, 190, 80–91. [CrossRef]
- 89. Benedek, M.; Kaernbach, C. Decomposition of skin conductance data by means of nonnegative deconvolution. *Psychophysiology* **2010**, *47*, 647–658. [CrossRef]
- 90. Fonseca, E.; Gong, R.; Bogdanov, D.; Slizovskaia, O.; Gómez Gutiérrez, E.; Serra, X. Acoustic scene classification by ensembling gradient boosting machine and convolutional neural networks. In Proceedings of theDetection and Classification of Acoustic Scenes and Events 2017 Workshop (DCASE2017), Munich, Germany, 16 November 2017; Virtanen, T., Mesaros, A., Heittola, T., Diment, A., Vincent, E., Benetos, E., Martinez, B., Eds.; Tampere University of Technology: Tampere, Finland, 2017; pp. 37–41.
- 91. Xu, K.; Zhu, B.; Kong, Q.; Mi, H.; Ding, B.; Wang, D.; Wang, H. General audio tagging with ensembling convolutional neural networks and statistical features. *J. Acoust. Soc. Am.* **2019**, *145*, EL521. [CrossRef]
- 92. Huppert, D.; Grill, E.; Brandt, T. A new questionnaire for estimating the severity of visual height intolerance and acrophobia by a metric interval scale. *Front. Neurol.* **2017**, *8*, 211. [CrossRef]

- 93. Balan, O.; Moise, G.; Moldoveanu, A.; Leordeanu, M.; Moldoveanu, F. Automatic Adaptation of Exposure Intensity in VR Acrophobia Therapy, Based on Deep Neural Networks. In Proceedings of the 27th European Conference on Information Systems—Information Systems for a Sharing Society, ECIS 2019, Stockholm, Sweden, 8–14 June 2019.
- 94. Bălan, O.; Moise, G.; Moldoveanu, A.; Leordeanu, M.; Moldoveanu, F. An investigation of various machine and deep learning techniques applied in automatic fear level detection and acrophobia virtual therapy. *Sensors* **2020**, *20*, 496.
- 95. Bălan, O.; Moise, G.; Moldoveanu, A.; Moldoveanu, F.; Leordeanu, M. Does automatic game difficulty level adjustment improve acrophobia therapy?: Differences from baseline. In Proceedings of the 24th ACM Symposium on Virtual Reality Software and Technology—VRST '18, Tokyo, Japan, 28 November–1 December 2018.
- 96. HTC Vive. Available online: https://www.vive.com/us/ (accessed on 20 September 2020).
- 97. Shimmer Sensing. Available online: https://www.shimmersensing.com/products/shimmer3-wireless-gsr-sensor (accessed on 20 September 2020).
- 98. Ledalab. Available online: http://www.ledalab.de/ (accessed on 20 September 2020).
- 99. Oskoei, M.A.; Hu, H. GA-based feature subset selection for myoelectric classification. In Proceedings of the 2006 IEEE International Conference on Robotics and Biomimetics, Kunming, China, 17–20 December 2006.
- 100. Granato, M.; Gadia, D.; Maggiorini, D.; Ripamonti, L.A. Feature extraction and selection for real-time emotion recognition in video games players. In Proceedings of the 2018 14th International Conference on Signal-Image Technology & Internet-Based Systems (SITIS), Las Palmas de Gran Canaria, Spain, 26–29 November 2018.
- 101. Chang, H.-A.; Chen, L.-F.; Chang, C.-C.; Tzeng, N.-S.; Kuo, T.J.; Kao, Y.-C.; Huang, S.-Y. Depression, anxiety, and heart rate variability: A case-control study in Taiwan. *J. Med Sci.* **2014**, *34*, 9. [CrossRef]
- 102. Zhang, X.; Wen, W.; Liu, G.; Hu, H. Recognition of public speaking anxiety on the recurrence quantification analysis of GSR signals. In Proceedings of the 2016 Sixth International Conference on Information Science and Technology (ICIST), Dalian, China, 6–8 May 2016; pp. 533–538.
- 103. Šalkevicius, J.; Damaševičius, R.; Maskeliunas, R.; Laukienė, I. Anxiety Level Recognition for Virtual Reality Therapy System Using Physiological Signals. *Electronics* 2019, *8*, 1039.
- 104. Sandulescu, V.; Andrews, S.; Ellis, D.; Bellotto, N.; Mozos, O.M. Stress detection using wearable physiological sensors. In Proceedings of the InternationalWork-Conference on the Interplay between Natural and Artificial Computation IWINAC 2015: Artificial Computation in Biology and Medicine, Elche, Spain, 1–5 June 2015; Springer: Cham, Switzerland, 2015; pp. 526–532.
- 105. Ihmig, F.R.; Neurohr-Parakenings, F.; Schäfer, S.K.; Lass-Hennemann, J.; Michael, T. On-line anxiety level detection from biosignals: Machine learning based on a randomized controlled trial with spider-fearful individuals. *PLoS ONE* **2020**, *15*, e0231517. [CrossRef] [PubMed]
- 106. Chen, L.-L.; Zhao, Y.; Ye, P.-F.; Zhang, J.; Zou, J.-Z. Detecting driving stress in physiological signals based on multimodal feature analysis and kernel classifiers. *Expert Syst. Appl.* **2017**, *85*, 279–291. [CrossRef]
- 107. Liu Y, Siqing D, Psychological Stress Level Detection Based on Electrodermal Activity. *Behav. Brain Res.* 2010. [CrossRef]
- 108. Sharma, V.; Prakash, N.R.; Kalra, P. EDA wavelet features as Social Anxiety Disorder (SAD) estimator in adolescent females. In Proceedings of the 2016 International Conference on Advances in Computing, Communications and Informatics (ICACCI), Jaipur, India, 21–24 September 2016; pp. 1843–1846. [CrossRef]

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



© 2020 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).